Mauna Loa Solar Observatory Observer's Log ______ Sun Jul 10 16:34:00 GMT 2005 Year: 05 Doy: 191 Observer: koon WEATHER COMMENT: Sun Jul 10 16:34:04 GMT 2005 Low clouds coming from the West, recent fog or rain, wind=5mph from the West, temp=44F. end COMMENT: Sun Jul 10 16:40:22 GMT 2005 Spar not tracking yet due to clouds. end A NEW PSPT EXABYTE IS LOADED ON KENOBI Sun Jul 10 16:40:59 GMT 2005 Sun Jul 10 16:48:12 GMT 2005 CHIP Start Patrol Sun Jul 10 16:48:22 GMT 2005 PICS Start Patrol A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Sun Jul 10 17:01:08 GMT 2005 Sun Jul 10 17:27:47 GMT 2005 CHIP End Patrol WEATHER COMMENT: Sun Jul 10 17:27:38 GMT 2005 Clouds are thick, fog upslope, pausing instruments and closing domes. end Sun Jul 10 17:28:17 GMT 2005 PICS End Patrol WEATHER COMMENT: Sun Jul 10 23:49:57 GMT 2005 Light rain now. end **GONG PROBLEM**: Sun Jul 10 23:50:04 GMT 2005 I installed the new HD as SCSI ID 2 for the workstation in the shelter, but the installation instructions weren't clear so I may have swapped the wrong drive, I'm waiting for a reply from GONGops on that. Meanwhile when I tried to restart the workstation I got filesystem error messages, I left it sitting and not completely restarted until I hear from GONGops about the drives. end **GONG PROBLEM**: Mon Jul 11 01:46:32 GMT 2005 I tried email and their pager but I haven't received a reply from GONGops regarding the HD swap problems. I don't want to risk problems by trying every possible swap combination so I'm going to put it back into the last known decent setup with the old HDs installed and the new HD on the workbench to be installed tomorrow. I'll do that on my way down today.

____end__

NICE PICSLIMB IMAGE: 1655 NICE CHIP IMAGE: 1654 NICE PICSDISC IMAGE: 1654 Mon Jul 11 01:56:24 GMT 2005

MkIV